Author's response to reviews

Title: Reproducibility and day time bias correction of optoelectronic leg volumetry: a prospective cohort study

Authors:

Rolf P Engelberger (rolf.engelberger@gmail.com)
Claudia Blazek (claudia.blazek@insel.ch)
Felix Amsler (felix.amsler@amslerconsulting.ch)
Hong H Keo (keoxx006@umn.edu)
Frédéric Baumann (frederic.baumann@insel.ch)
Werner Blättler (w.blaettler@gmail.com)
Iris Baumgartner (iris.baumgartner@insel.ch)
Torsten Willenberg (torsten.willenberg@insel.ch)

Version: 3 Date: 2 October 2011

Author's response to reviews: see over
Dear Editor,

Thank you very much for accepting our manuscript. We have performed the format changes according to your suggestions. Please find our comments below.

Major revisions

Abstract: This should be identical in the manuscript and the submission system

The abstract in the submission system has been updated accordingly.

Ethical approval: Any experimental research that is reported in the manuscript should have been performed with the approval of an appropriate ethics committee. The statement to this effect must include the name of the body which gave approval, with a reference number where appropriate.

The name of the local ethics committee has been added under the Method section

Minor revisions

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

This format change has been performed accordingly.

Please remove word counts and line numbers

Both modifications have been done.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Both figures have been cropped according to your online figure guide and uploaded as PowerPoint files.

Tables: the table title should appear above the table and should be only one sentence
long. Please place any other text as a legend (or key), which should appear below the table

For the three tables, the title and the note have been separated from the table itself and have been added above and below the corresponding table.

Please present Table 2 in landscape format. At present, it is cut off by the edge of the page.

This format change has been performed.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have checked our manuscript for typographical errors.

Sincerely,

Rolf Engelberger, MD